Author's response to reviews

Title: Processes, Contexts and Rationale for Disinvestment: A Protocol for a Critical Interpretive Synthesis

Authors:

Michael G Wilson (wilsom2@mcmaster.ca)
Moriah E Ellen (moriah.ellen@gmail.com)
John N Lavis (lavisj@mcmaster.ca)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Kaelan A Moat (moatka@mcmaster.ca)
Joshua Shemer (shemer@assuta.com)
Terry Sullivan (tsulliva2@gmail.com)
Sarah Garner (Sarah.Garner@nice.org.uk)
Ron Goeree (goereer@mcmaster.ca)
Roberto Grilli (RGrilli@Regione.Emilia-Romagna.it)
Justin Peffer (Justin.Peffer@ontario.ca)
Kevin Samra (kevin.samra@gov.bc.ca)

Version: 4 Date: 10 November 2014

Author's response to reviews: see over
Dear Systematic Reviews editors,

On behalf of all the authors, I would like to thank you for accepting our manuscript for publication in Systematic Reviews. In response to requests made by the editors, we have moved the key words and PROSPERO registration from the title page to beneath the abstract and added the information related to the additional files to below the reference list. We did not make any changes to the author acknowledgements as our existing content seems to already comply with your suggestion. The sentence indicates that all authors provided input into the original grant proposal to the Canadian Institutes of Health Research and have read and approved this manuscript. The initials of each author are not listed but the example provided also didn’t list initials for a similar phrase when referring to all authors.

Thank you again for publishing our manuscript in Systematic Reviews.

Sincerely,

Michael G. Wilson, PhD

Assistant Director, McMaster Health Forum
Assistant Professor, Department of Clinical Epidemiology and Biostatistics
Member, Centre for Health Economics and Policy Analysis
McMaster University
1280 Main St. West, CRL-223
Hamilton, ON, Canada L8S 4K1
Tel: +1 905.525.9140 ext. 22121
Email: wilsom2@mcmaster.ca